Author's response to reviews

Title: Metals and kidney markers in adult offspring of endemic nephropathy patients and controls. A two-year follow-up study.

Authors:

Wilfried Karmaus (karmaus@sc.edu)
Plamen Dimitrov (p.dimitrov@ncphp.government.bg)
Valeri Simeonov (p.dimitrov@ncphp.government.bg)
Svetla Tzolova (sv.tzolova@ncphp.government.bg)
Angel Bonev (p.dimitrov@ncphp.government.bg)
Rossitza Georgieva (r.georgieva@ncphp.government.bg)

Version: 4 Date: 19 February 2008

Author's response to reviews:

Dear Environmental Health Editorial Team:
Please find the paper with an updated (revised) title.
Thanks
Wilfried Karmaus